Author's response to reviews

Title: Unpredicted spontaneous extrusion of a renal calculus in an adult male with spina bifida and paraplegia: report of a misdiagnosis. Measures to be taken to reduce urological errors in spinal cord injury patients

Authors:

Subramanian Vaidyanathan (vaidyanathansiu@hotmail.com)
Peter L Hughes (Peter.Hughes@mail.soh-tr.nwest.nhs.uk)
Bakul M Soni (Bakul.Soni@mail.soh-tr.nwest.nhs.uk)
Gurpreet Singh (gurpreet.singh@ukgateway.net)
Paul Mansour (Paul.Mansour@mail.soh-tr.nwest.nhs.uk)
Pradipkumar Sett (Pradip.Sett@mail.soh-tr.nwest.nhs.uk)

Version: 9 Date: 18 Dec 2001

Dear Editor
Thank you for considering our submission for publication in your esteemed journal. We changed the title as advised by the referee. We deleted the section “What do we learn from this case?”. We omitted one reference pertaining to this section. We changed the word “medical” to “urological” in the manuscript. We mentioned that we did not perform ultrasound scan of loin wall. We accepted our folly that we did not carry out CT scan of the abdomen and stated explicitly our over reliance on ultrasound scan. We stated that blood test was not performed. Typographic error on page 6, line 4 from the top, has been corrected.
I am grateful to you for your valuable help.
I wish you Merry Christmas.
Dr S Vaidyanathan